Author's response to reviews

Title: Evaluation of Endoscopic Vein Extraction on Structural and Functional Viability of Saphenous Vein Endothelium

Authors:

Bader E. Hussaini (bcell@hotmail.com)
Xiu-Gui Lu (xiu-gui_lu@hms.harvard.edu)
J. Alan Wolfe (heartjock@yahoo.com)
Hemant S. Thatte (hemant_thatte@hms.harvard.edu)

Version: 4 Date: 7 June 2011

Author's response to reviews: see over
June 06, 2011

Editorial Office
Journal of Cardiothoracic Surgery

Re: Manuscript ID: 1453676130534388

Title: Evaluation of Endoscopic Vein Extraction on Structural and Functional Viability of Saphenous Vein Endothelium

Dear Editors:

Enclosed is the resubmission of the above manuscript after revision. Two sections have been added to the manuscript, the “Competing Interests’ and “ Authors Contribution”. These two items were the only formatting changes that were requested by the publisher.

We hope this revised manuscript is more acceptable for publication in the Journal of Cardiothoracic Surgery.

Sincerely;

Bader E. Hussaini

Bader E. Hussaini, MD, MPH
Instructor in Surgery
Cardiac Surgery Department
VA Boston Healthcare System
Brigham and Women’s Hospital
Harvard Medical School